Author's response to reviews

Title: The natural triterpene maslinic acid induces apoptosis in HT29 colon cancer cells by a JNK-p53-dependent mechanism.

Authors:

Fernando J Reyes-Zurita (ferjes@ugr.es)
Gisela Pachón-Peña (giselapachon@ub.edu)
Daneida Lizárraga (D.Lizarraga@grat.unimaas.nl)
Eva E Rufino-Palomares (evaevae@ugr.es)
Marta Cascante (martacascante@ub.edu)
José A Lupiáñez (jlcara@ugr.es)

Version: 4 Date: 7 March 2011

Author's response to reviews: see over
Dear Professor Neilan,

Please find enclosed our revised manuscript entitled “The natural triterpene maslinic acid induces apoptosis in HT29 colon cancer cells by a JNK-p53-dependent mechanism”. According with yours last conditions, we have modified the abstract structure including all the recommended sections. With this, we hope that you will finally find our paper suitable for publication in BMC-Cancer.

Thank you very much for your time and consideration. I look forward to hearing you soon.

Yours sincerely,

[Signature]

Prof. Dr. José A. Lupiáñez
Department of Biochemistry and Molecular Biology I
University of Granada
Campus Fuentenueva
18071 Granada
Spain
Tel. +34 958 243089, Fax: 34 958 249945, E-mail: jlcara@ugr.es